Author’s response to reviews

Title: Successful treatment with anti-programmed-death-1 antibody in a relapsed natural killer/T-cell lymphoma patient with multi-line resistance: A case report

Authors:

Jianping Lai (yipingqingyun@163.com)
Peng Xu (512073602@qq.com)
Xiaoliu Jiang (923165950@qq.com)
Shan Zhou (259324509@qq.com)
Anwen Liu (awliu666@163.com)

Version: 2 Date: 16 Jun 2017

Author’s response to reviews:

Dear Catherine Rice,

On behalf of my co-authors, we thank you very much for giving us an opportunity to revise our manuscript, we appreciate editor and reviewers very much for their positive and constructive comments and suggestions on our revised manuscript entitled “Successful treatment with anti-PD-1 antibody in a relapsed NK/T lymphoma patient with multi-line resistance: a case report”. (BCAN-D-17-00424R1).

We have studied reviewer’s comments carefully and have tried our best to revise our manuscript according to the comments. Attached please find the revised version, which we would like to submit for your kind consideration.

We would like to express our great appreciation to you and reviewers for comments on our paper. Looking forward to hearing from you.

Thank you and best regards.

Yours sincerely,

Jianping Lai
Dear Editors and Reviewers:

Thank you for your letter and for the reviewers’ comments concerning our revised manuscript entitled “Successful treatment with anti-PD-1 antibody in a relapsed NK/T lymphoma patient with multi-line resistance: a case report”. (BCAN-D-17-00424R1). Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. We have studied comments carefully and have made correction which we hope meet with approval. The main corrections in the paper and the responds to the reviewer’s comments are as flowing:

Responds to the editor Comments:

1. Response to comment: Please provide email address details for all authors on the title page of the manuscript.
Response: Email address details were provided for all authors on the title page of the manuscript.

2. Response to comment: Please ensure that there are specific authors' contributions for all of the authors.
Response: Specific authors' contributions were ensured for all of the authors in Authors’ contributions section.

3. Response to comment: Please check our authorship criteria and ensure all named authors have provided sufficient input to be so registered.
Response: We have checked our authorship criteria and ensured all named authors have provided sufficient input to be so registered.
4. Response to comment: Figures should be provided as separate files, and each figure of a manuscript should be submitted as a single file.

Response: each figure of a manuscript was submitted as a single file.

5. Response to comment: Please check your Discussions and conclusions section to remove text overlap with other previously published works.

Response: We have rewritten Discussions and conclusions section.

6. Response to comment: We recommend that you ask a native English speaking colleague to help you copyedit the paper.

Response: The paper was copyedited by a native English speaking colleague.

7. Response to comment: At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethrough or text in different colours. All relevant tables and figures should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: We resubmitted the revised manuscript and figures as you asked.

Responds to the reviewer’s comments:

Reviewer #1 (Enrico Derenzini):

1. Response to comment: The main limitation of this manuscript is still the quality of written english, which needs to be carefully revised.

Response: manuscript was reviewed by a native English speaking colleague who is fluent in English.

Special thanks to you for your good comments.

Yours sincerely,

Jianpig Lai
Corresponding author:

Name: Anwen Liu

E-mail: awliu666@163.com